ISOLDE Yes Beam Availability ISOLDE 98.6% Facil Summary All operational beams • AWAKE and VdM bea	Scheduled PS Yes / by Destination (AFT) PS 98.5% ity Status delivered as requested.
Beam ISOLDE Yes Beam Availability ISOLDE 98.6% Facil Summary All operational beams • AWAKE and VdM beam	Scheduled Yes PS Yes py Destination (AFT) 98.5% Ity Status 98.5% ity Status 98.5% odelivered as requested. 98.5% ams ready 98.5% o with missing current acquisition in IPOC for ring 3 ably a card has to be replaced) 98.5% fluctuations: two accesses required. A card was 98.5%
ISOLDE Yes Beam Availability ISOLDE 98.6% Facil Summary All operational beams • AWAKE and VdM bea	PS Yes v by Destination (AFT) PS PS 98.5% ity Status 98.5% o delivered as requested. 98.5% ams ready on with missing current acquisition in IPOC for ring 3 ably a card has to be replaced) fluctuations: two accesses required. A card was
Beam Availability ISOLDE 98.6% Facil Summary • All operational beams • AWAKE and VdM beams	Image: View of the state o
ISOLDE 98.6% Facil Summary All operational beams • AWAKE and VdM beams	PS 98.5% ity Status 98.5% </th
Summary All operational beams • AWAKE and VdM beams	ity Status delivered as requested. ams ready with missing current acquisition in IPOC for ring 3 ably a card has to be replaced) fluctuations: two accesses required. A card was
Summary • All operational beams • AWAKE and VdM beams	a delivered as requested. ams ready with missing current acquisition in IPOC for ring 3 ably a card has to be replaced) fluctuations: two accesses required. A card was
AWAKE and VdM bea	ams ready with missing current acquisition in IPOC for ring 3 ably a card has to be replaced) fluctuations: two accesses required. A card was
 Non-blocking problem 	ably a card has to be replaced) fluctuations: two accesses required. A card was
 KSWs persists (probation in the second second	ally solved the situation Int polarity requested for some cycles but not ctromechanical switch) ==> set 0.5 A to all users HRS and NORMHRS poling: the alarm was not published in the on LASER running on CWO-CCC-B0LF. The ged when opening another session on CWO- of the cooling control system will be rebooted at the problem is probably on the LASER side (to be gains are no more automatically set by the cruise It values, found by Jose, are set for the different 4db for AD, MTE, TOF, ISOGPS/HRS type Odb for all LHC, STAGISO beams. converter exchanged on 09/09 morning
Plans Deliver beams to downstream facilities as needed Intervention Request	
No Duration	Preferred date/time
Reason	
Impact	